

Global Biogeochemical Cycles[®]

RESEARCH ARTICLE

10.1029/2023GB008039

Key Points:

- Decreased hydrodynamics and water depth created a favorable environment for eelgrass establishment about 2,000 years ago
- Carbon and nitrogen burial increased in order of magnitudes following seagrass colonization
- Paleoecological information on environmental conditions linked to seagrass colonization can aid conservation and restoration efforts

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

M. Dahl, martin.dahl@sh.se

Citation:

Dahl, M., Gullström, M., Bernabeu, I., Serrano, O., Leiva-Dueñas, C., Linderholm, H. W., et al. (2024). A 2,000year record of eelgrass (*Zostera marina* L.) colonization shows substantial gains in blue carbon storage and nutrient retention. *Global Biogeochemical Cycles*, *38*, e2023GB008039. https://doi.org/10.1029/ 2023GB008039

Received 13 NOV 2023 Accepted 9 FEB 2024

Author Contributions:

Conceptualization: Martin Dahl, Martin Gullström, Maria E. Asplund, Mats Björk, J. Robin Svensson, Miguel A. Mateo Formal analysis: Martin Dahl, Irene Bernabeu, Oscar Serrano, Carmen Leiva-Dueñas, Hans W. Linderholm, Tinghai Ou, Elinor Andrén, Thomas Andrén, Sanne Bergman, Sara Braun, Anneli Eklöf, Zilvinas Ežerinskis, Andrius Garbaras, Petter Hällberg, Elin Löfgren, Malin E. Kylander, Pere Masqué, Justina Šapolaitė, Rienk Smittenberg, Miguel A. Mateo

© 2024. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

A 2,000-Year Record of Eelgrass (*Zostera marina* L.) Colonization Shows Substantial Gains in Blue Carbon Storage and Nutrient Retention

Martin Dahl¹ ^(D), Martin Gullström¹ ^(D), Irene Bernabeu² ^(D), Oscar Serrano^{2,3} ^(D), Carmen Leiva-Dueñas⁴ ^(D), Hans W. Linderholm⁵ ^(D), Maria E. Asplund⁶ ^(D), Mats Björk⁷ ^(D), Tinghai Ou⁵ ^(D), J. Robin Svensson⁸ ^(D), Elinor Andrén¹ ^(D), Thomas Andrén¹ ^(D), Sanne Bergman⁹ ^(D), Sara Braun¹ ^(D), Anneli Eklöf⁷, Zilvinas Ežerinskis¹⁰, Andrius Garbaras¹⁰ ^(D), Petter Hällberg¹¹ ^(D), Elin Löfgren⁷, Malin E. Kylander¹² ^(D), Pere Masqué^{3,13} ^(D), Justina Šapolaitė¹⁰, Rienk Smittenberg^{11,14} ^(D), and Miguel A. Mateo^{2,3}

¹School of Natural Sciences, Technology and Environmental Studies, Södertörn University, Huddinge, Sweden, ²Centro de Estudios Avanzados de Blanes, Consejo Superior de Investigaciones Científicas (CEAB-CSIC), Blanes, Spain, ³Centre for Marine Ecosystems Research, School of Natural Sciences, Edith Cowan University, Joondalup, WA, Australia, ⁴Department of Ecoscience, Aarhus University, Aarhus C, Denmark, ⁵Regional Climate Group, Department of Earth Sciences, University of Gothenburg, Gothenburg, Sweden, ⁶Department of Biological and Environmental Sciences, University of Gothenburg, Kristineberg, Fiskebäckskil, Sweden, ⁷Department of Ecology, Environment and Plant Sciences, Stockholm University, Stockholm, Sweden, ⁸Department of Marine Sciences, University of Gothenburg, Gothenburg, Sweden, ⁹The Arctic University Museum of Norway, UiT—The Arctic University of Norway, Tromsø, Norway, ¹⁰Center for Physical Sciences and Technology, Vilnius, Lithuania, ¹¹Department of Geological Sciences, Stockholm University, Stockholm, Sweden, ¹³International Atomic Energy Agency, Principality of Monaco, Monaco, ¹⁴Swiss Federal Institute for Forest, Snow and Landscape Research WSL, Birmensdorf, Switzerland

Abstract Assessing historical environmental conditions linked to habitat colonization is important for understanding long-term resilience and improving conservation and restoration efforts. Such information is lacking for the seagrass Zostera marina, an important foundation species across cold-temperate coastal areas of the Northern Hemisphere. Here, we reconstructed environmental conditions during the last 14,000 years from sediment cores in two eelgrass (Z. marina) meadows along the Swedish west coast, with the main aims to identify the time frame of seagrass colonization and describe subsequent biogeochemical changes following establishment. Based on vegetation proxies (lipid biomarkers), eelgrass colonization occurred about 2,000 years ago after geomorphological changes that resulted in a shallow, sheltered environment favoring seagrass growth. Seagrass establishment led to up to 20- and 24-fold increases in sedimentary carbon and nitrogen accumulation rates, respectively. This demonstrates the capacity of seagrasses as efficient ecosystem engineers and their role in global change mitigation and adaptation through CO₂ removal, and nutrient and sediment retention. By combining regional climate projections and landscape models, we assessed potential climate change effects on seagrass growth, productivity and distribution until 2100. These predictions showed that seagrass meadows are mostly at risk from increased sedimentation and hydrodynamic changes, while the impact from sea level rise alone might be of less importance in the studied area. This study showcases the positive feedback between seagrass colonization and environmental conditions, which holds promise for successful conservation and restoration efforts aimed at supporting climate change mitigation and adaptation, and the provision of several other crucial ecosystem services.

Plain Language Summary This study investigated the historical colonization of eelgrass (*Zostera marina*), an important marine vascular plant in cold-temperate coastal regions. Sediment cores from eelgrass meadows on the Swedish west coast dating back as far as 14,000 years were examined to understand the time-course of eelgrass colonization and the subsequent modification of the environment. We found that eelgrass colonization began approximately 2,000 years ago, coinciding with the development of shallow, sheltered conditions that favored eelgrass growth. As eelgrass became established, this led to substantial habitat and sediment changes, including a 20- and 24-fold increase in carbon and nitrogen accumulation, respectively. This highlights the crucial role of eelgrass as a provider of important ecosystem services, such as regulation of climate, nutrient retention, and sediment protection. We also examined the potential effects of climate change on eelgrass growth and health, predicting that decreased water clarity and altered water flow pose the greatest risks.



Funding acquisition: Martin Dahl, Martin Gullström, Oscar Serrano,

Martin Gullström, Maria E. Asplund, Mats Björk, J. Robin Svensson Writing – original draft: Martin Dahl Writing – review & editing: Martin Gullström, Irene Bernabeu, Oscar Serrano, Carmen Leiva-Dueñas,

Hans W. Linderholm, Maria E. Asplund,

J. Robin Svensson, Elinor Andrén, Thomas Andrén, Sanne Bergman,

Zilvinas Ežerinskis, Andrius Garbaras,

Petter Hällberg, Elin Löfgren, Malin E. Kylander, Pere Masqué,

Justina Šapolaitė, Rienk Smittenberg,

Mats Björk, Miguel A. Mateo

Investigation: Martin Dahl,

Mats Biörk, Tinghai Ou.

Sara Braun, Anneli Eklöf,

Miguel A. Mateo

Overall, this study adds valuable insights into the relationship between eelgrass and its environment, aiding in conservation and restoration efforts to mitigate climate change and maintain essential ecosystem services. It emphasizes the importance of specific environmental conditions for successful eelgrass colonization and restoration.

1. Introduction

Coastal vegetated habitats, including seagrass meadows, are highly productive and diverse marine ecosystems (Barbier et al., 2011). Seagrass meadows provide a myriad of ecosystem services, such as biodiversity maintenance, provision of nursery habitats, contribution to climate change mitigation and coastal protection, and water purification (Cullen-Unsworth & Unsworth, 2013; de los Santos et al., 2020; Macreadie et al., 2021). Thus, seagrasses contribute important ecosystem services for the wellbeing of people and the planet. Through the reduction of hydrodynamic forces within the seagrass canopy and stabilization of sediment within the rootrhizome system, seagrasses trap and embed organic and inorganic particles (Lei et al., 2023; Samper-Villarreal et al., 2016) of both allochthonous and autochthonous origin (Asplund et al., 2021; Kennedy et al., 2010; Oreska et al., 2018) leading to accumulation of thick organic-rich sediment deposits, which can remain stable over millennia (Mateo et al., 1997). Through continuous sediment accretion, seagrass meadows support long-term blue carbon storage, nutrient retention and reduced turbidity (through sediment stabilization) in coastal areas (Lima et al., 2020; Mazarrasa et al., 2018).

Eelgrass (Zostera marina L.) is one of the most prevalent seagrass species globally with a geographical distribution spanning the entire cold-temperate zone of the Northern hemisphere (Short et al., 2007). The colonization of seagrasses in cold-temperate regions is highly dependent on the attenuation of irradiance with increasing water depth, and the lower depth limit of Z. marina distribution is usually regulated by water clarity and turbidity (Krause-Jensen et al., 2011; Nielsen et al., 2002), which is often negatively impacted by eutrophication, coastal exploitation (e.g., sedimentation) and climate change (through increased precipitation and land runoff) (Rabalais et al., 2009). The colonization, growth and distribution of Z. marina are governed by environmental conditions, such as hydrodynamic-driven turbidity and water depth, and human-induced disturbances, affecting water quality. To understand the historical establishment and subsequent colonization of seagrass plants, paleoreconstruction of sedimentary records has been successfully performed for seagrass meadows (Serrano et al., 2020), providing insights into the dynamics of ecosystem health status in response to long-term environmental change (Leiva-Dueñas et al., 2021; López-Sáez et al., 2009; Mateo et al., 2010). As seagrass loss and recolonization is related to environmental dynamics, information on past environmental conditions linked to seagrass ecosystem integrity could be of vital help for coastal managers guiding seagrass restoration efforts as well as to predict impacts on seagrass distribution and provisioning of ecosystem services from future climate change.

Climate change is considered a major threat to seagrass habitats (Short & Neckles, 1999), potentially leading to dramatic changes in the coastal environment, including loss of seagrass meadows (Orth et al., 2006). Global temperatures have drastically increased during the twenty-first century (IPCC, 2021), which entails the thermal expansion of the oceans and melting of land ice. Therefore, sea levels are predicted to increase globally (Masson-Delmotte et al., 2021), although regionally the rising sea levels could offset local sea level regressions from ongoing isostatic uplift (Meier et al., 2004). In addition, precipitation in the cold-temperate zone has increased and even higher precipitation is expected, with stronger rainfall events in the near future (Pörtner et al., 2022), which can lead to decreased salinity in coastal waters due to higher freshwater runoff from land (Cheng et al., 2020) that may in turn negatively influence seagrass distribution (Stevens & Lacy, 2012).

For cold-temperate seagrass species, such as *Z. marina*, using paleoreconstructions for describing past environmental conditions and to predict future changes is lacking (but see Kindeberg et al., 2019; Leiva-Dueñas et al., 2023). There are a number of seagrass-directed paleorecords of the genus *Posidonia* (López-Merino et al., 2017; Macreadie et al., 2012; Mateo et al., 1997, 2010; Serrano et al., 2011, 2016), but these are geographically constrained to Australia and the Mediterranean. Therefore, understanding the establishment and colonization of the widespread *Z. marina* is of high relevance and can likely apply to other seagrass species growing under similar environmental conditions. Here we aim to reconstruct millenary changes in habitat



condition in response to environmental changes based on biogeochemical analyses of two sediment cores from *Z. marina* meadows to (a) establish a time frame for seagrass colonization, (b) assess the climatic and geomorphological conditions that favored seagrass colonization, (c) describe the biogeochemical changes in the sediment following the establishment of the seagrass meadows, and (d) predict potential future climate change impacts (from sea level rise [SLR], decrease in salinity and increase in runoff and sea surface temperature [SST]) on the seagrass meadow health status.

2. Methods

2.1. Study Context

The Swedish coastal zone has been severely impacted by eutrophication and sediment supply from land over the last century, resulting in a recession in the depth limit distribution of *Z. marina* (Boström et al., 2014; Rosenberg et al., 1996). Historical reports for *Z. marina* on the Swedish west coast showed depths of 15–20 m (Loo & Isaksson, 2015; Petersen, 1893), while today eelgrass plants are rarely found below 9 m (Skåne County Administrative Board, 2016; pers. obs. Asplund and Svensson) and generally growing in water depths from 0.5 to 6 m (Svensson et al., 2021). In addition to light conditions and turbidity levels, seawater temperature and salinity can negatively affect *Z. marina* performance (Marsh et al., 1986; Rasmusson et al., 2020; Xu et al., 2016).

Along the Swedish coast, fluctuations in sea level, temperature and salinity have occurred over the Holocene (the last ~11.7 cal ka BP) due to isostatic and eustatic changes following the retreat of the Scandinavian ice sheet (Berner et al., 2011; Mörner, 1969; Törnqvist & Hijma, 2012) leading to geomorphological transformations along the coastline. Warming in the early Holocene caused the ice sheet to retreat and the westward outlet opened the drainage of the Baltic ice lake at 11.7 cal ka BP (Andrén et al., 2011; Björck, 2008; Jakobsson et al., 2007). This was followed by a series of dynamic changes in hydrographic conditions affecting sedimentation (Gyllencreutz et al., 2006) and at about 8.5 cal ka BP, there was a shift in the ocean circulation resulting in a higher inflow of marine waters (Conradsen & Heier-Nielsen, 1995). During the last ~2 cal ka BP, the Swedish Skagerrak coast has been dominated by Atlantic water conditions (Erbs-Hansen et al., 2012). Periods of increased temperatures occurred during the Roman Warm Period (RWP) (~2.3 to 1.6 cal ka BP) (Hass, 1996; Neukom et al., 2019) and the Medieval Climate Anomaly (MCA) (~1.2 to 0.9 cal ka BP) (Mann et al., 2009), and these warmer periods were linked to less frequent storms and lower sedimentation runoff in the Skagerrak region (Hass, 1996).

2.2. Study Area and Sediment Core Sampling

Field sampling was conducted in two Z. marina meadows at Gåsö (58°14'10"N, 11°23'40"E), situated in the vicinity of the Gullmar Fjord mouth on the Swedish Skagerrak coast (Figure 1a), in August 2020. Gåsö is a small island ($\sim 3 \text{ km}^2$) mainly composed of granitic bedrock (Figure 1b). In the middle of the island, connecting the western and the eastern parts, there is a low-elevation (less than 1 m above sea level [m a. s. l.]) postglacial sand deposit. The island has a relatively low exposure to human activities with only a few country residences in the eastern part. On the southern coast of the island, there is an up to 3 m deep elongated cove stretching toward the center of the island, whereas on the northern side there is a shallow (up to 4 m depth) embayment surrounded by smaller rocks and islets creating a hydrodynamically sheltered environment. The main wind directions are from the W and SW. Within both the northern and southern embayments, seagrass meadows are found growing from about 0.5 to 4 m water depth. The distribution of eelgrass was determined by encircling the meadows by boat repeatedly crossing the edge of the meadow with support of sonar and a sub-surface camera for validation of Z. marina vegetation. GPS waypoints were collected at the intersections and corrected for geometry errors caused by drifting prior to vectorization. The mapped seagrass area was approximately 7 and 5 ha at Gåsö S and N, respectively. In the middle of each of the seagrass meadows at Gåsö South (S) and Gåsö North (N), a sediment core (2 m long and 7.5 cm in diameter) was collected using SCUBA by hammering a PVC-corer. The sediment cores in Gåsö S and N were collected at 1.8 and 3 m water depths, respectively (Figure 1c), and the top \sim 50 cm of the sediment cores have previously been described by Dahl et al. (2023). Sediment core compression was assessed once during coring by measuring the inner and outer portions of the corer before core extrusion from the seagrass meadow. From these values, a compression factor was calculated, which was similar for the two cores (18% and 17% in Gåsö S and N, respectively). The compression factors were applied to sediment depth and dry bulk density (DBD) to calculate decompressed values, which were used in subsequent analysis. The cores were stored in a cold





Figure 1. Maps showing (a) the study locations (Gåsö South [S] and Gåsö North [N]) in the vicinity of the Gullmar Fjord, (b) substrate types at Gåsö, and (c) water depth ranging from 0 to 6 m and current distribution of eelgrass.

room and were cut lengthwise into two hemi-cores. One half was sliced into 1 cm intervals and the other half was kept intact for X-ray fluorescence (XRF) measurements.

2.3. Sediment Chronology

The concentrations of ²¹⁰Pb were analyzed by measuring its decay product (²¹⁰Po) in equilibrium using alpha spectrometry (Sanchez-Cabeza et al., 1998), which aimed to obtain age models for the two cores. Details of the ²¹⁰Pb analysis are presented in Dahl et al. (2023). Ten radiocarbon (¹⁴C) dates were obtained along the cores (6 in Gåsö N and 4 in Gåsö S). Prior to ¹⁴C measurements in a single-stage Accelerator Mass Spectrometer (AMS), samples were digested using the standard acid-base-acid method (Molnár et al., 2013). The background values of measurements were estimated to be 0.25 pMC using phthalic anhydride with the NIST-OXII (134.06 pMC) standard used as reference material. The ¹⁴C/¹²C ratio was measured with an accuracy of >0.3%. For the isotopic fractionation correction, the ratio of ¹³C/¹²C was used. The ¹⁴C ages were calibrated using the Marine20 calibration curve (Heaton et al., 2020) and corrected for a local marine reservoir effect ($\Delta R = -208 \pm 57$ years) based

on map nos. 77, 681–684, and 674 (Håkansson, 1970, 1987; Olsson, 1980) in the Marine20 reservoir database. Core age models based on both ²¹⁰Pb- and ¹⁴C-dates were estimated using the R-package Bacon (Blaauw & Christeny, 2011). The sediment accumulation rates (SAR) and mass accumulation rates (MAR) were calculated using the Constant Rate of Supply (CRS) models for the top ²¹⁰Pb-dated sediment cores, and from the Bacon models based on cumulative mass versus age (Belshe et al., 2019) for the older sediment layers. SAR was later calculated by dividing the Bacon-derived MAR by the sediment DBD.

2.4. Sediment Biogeochemical Analyses

All sediment slices (n = 166 and 148 for Gåsö N and S, respectively) were dried at 60°C until constant weight was obtained to estimate DBD (g cm⁻³) (see data set [DOI: 10.5281/zenodo.10559287] for details on biogeochemical analysis performed along the cores). Samples (n = 21 and 22 for Gåsö N and S, respectively) for organic carbon (OC) and nitrogen (N) contents and C and N stable isotopic composition (δ^{13} C and δ^{15} N) were analyzed using a Carlo Erba NC2500 elemental analyzer connected to a Thermo Scientific Delta V Advantage Isotope Ratio Mass Spectrometer (EA-IRMS). Acetanilid ($\delta^{13}C = -26.14 \pm 0.15\%$, $\delta^{15}N = 0.38 \pm 0.12\%$) was used as the reference material and the standard deviation was approximately 1% for C and N concentrations and 0.1% for both C and N stable isotope ratios. The isotopic compositions were expressed in delta notation (per mil) relative to the VPDB (Vienna PeeDee Belemnite) for δ^{13} C and to the atmospheric nitrogen standard for δ^{15} N. Prior to the OC and N analysis, the sediment was ground using a mixer mill (Retsch MM 400), weighted in silver capsules and treated with 1 M HCl to remove inorganic carbon (direct addition using a pipet) (Dahl et al., 2016). Carbon accumulation rates (CAR) and nitrogen accumulation rates (NAR) were calculated using the weighted mean of OC and N contents and MAR (Arias-Ortiz et al., 2020). Calcium carbonate (CaCO₃) contents were estimated from an aliquot of dry sediment (n = 28 and 14 for Gåsö N and S, respectively) through loss on ignition combustion at 450°C for 6 hr followed by 2 hr at 950°C following Bengtsson and Enell (1986) and Heiri et al. (2001). Grain size distribution following the classification of Wentworth (1922) was performed using a laser diffraction particle size analyzer (Mastersizer 2000 MALVERN). Prior to analysis, the samples (n = 28 and 24 for Gåsö N and S, respectively) were sieved by 2 mm and the fraction <2 mm was treated with 30% H₂O₂ to remove organic matter. Mean grain size (µm) and sorting coefficient were calculated using the Gradistats program (Blott & Pye, 2001) based on Folk and Ward (1957).

Lipid analysis was carried out to trace seagrass plant-derived material in the sediment record (Jaffé et al., 2001; Sikes et al., 2009). For the analysis, 14 samples from each core as well as above- (n = 3) and belowground biomass (n = 3) of seagrass were freeze-dried and milled. A mixture of dichloromethane (DCM) and methanol (MeOH) (9:1 v/v) was added to the dried sediment, and samples were placed in a sonic bath for 15 min (Poynter & Eglinton, 1990). The samples were then centrifuged at 800 rpm for 10 min and the lipid extract was placed in a glass tube. This was repeated three times and the lipid extracts were combined. Activated copper powder was added to the lipid extracts for removal of elemental sulfur, and the lipid extracts were dried under a N₂ blowdown system and later re-dissolved in <1 mL DCM. Deactivated 95% silica gel was added to adsorb the lipid extract and the adsorbed samples were placed in glass pipettes that had been packed with pre-combusted (400°C, overnight) and deactivated (addition of 5% H₂O by weight) silica gel, and the fractions of non-polar hydrocarbons were eluted using hexane. These fractions were dissolved in 400-1,000 µL hexane depending on the expected concentration on a Shimadzu QP2010 Ultra GC-MS equipped with an AOC-20i auto sampler and a split-splitless injector operated in splitless mode. A Zebron ZB-5HT Inferno GC column (30 m \times 0.25 mm \times 0.25 µm) was used for separation. An external standard containing a mixture of $C_{20}-C_{40}$ n-alkanes with known concentration was analyzed in conjunction with the samples and used for quantification based on peak areas. The average chain length (ACL) (for homologs between C₁₇ and C₃₅) was calculated following ACL = $\sum (n \times C_n) / \sum C_n$, where *n* is the number of carbon atoms and C_n is the concentration (mg g DW⁻¹) of the *n*-alkane.

Magnetic susceptibility (MS) was used as a proxy to assess the abundance of ferromagnetic minerals derived from terrestrial soil erosion to assess the relative relationship between marine and terrestrial influences on sedimentation (López-Merino et al., 2017). MS was analyzed in 37 and 40 samples from Gåsö N and S, respectively, using a Multi Sensor Core Logger (LABCORE, University of Barcelona). Samples were packed in a 7 cm³ cubicle and analyzed with a low frequency (~0.1 mT) volumetric MS. Each sample was analyzed five times and calibration was manually performed before starting each measurement. Mass specific MS (χ ; cm³ g⁻¹) was calculated as $\chi = \kappa / \rho$, where κ is the average of the replicates and ρ is sample density (cm³ g⁻¹) (Hatfield et al., 2013).



XRF measurements were performed on the intact hemi-cores using an ITRAX XRF core scanner from Cox Analytical Systems, which produces digital imagery of the cores and μ -XRF elemental profiles (Si, S, Cl, K, Ca, Ti, Mn, Fe, Ni, Cu, Zn, Br, Rb, Sr, Zr and Pb). The core scanner used a Mo tube with the setting of 30 kV and 50 mA, and a step size of 500 µm with a dwell time of 25 s. Before data handling, the μ -XRF elemental profiles were normalized using centered log-ratio transformation, which also accounts for non-linear effects of the elemental matrix (Weltje & Tjallingii, 2008).

2.5. Climate Modeling

Future changes in sea level were extracted from the IPCC AR6 SLR Projections (Fox-Kemper et al., 2021; Garner et al., 2021). The medium-confidence projections of the integrated SLR over all components, including the Antarctic and Greenland ice sheets, glaciers, land water storage, ocean dynamics (including thermal expansion), and vertical land motion (non-climatic processes), were utilized. Although the projections were available for the period CE 2020–2150 at a 10-year resolution, with the baseline period of 1995–2014, we limited our projection to 2100. Historical simulations and future projections of SST, sea surface salinity, and surface runoff were derived from global climate models (GCMs) in the Coupled Model Intercomparison Project Phase 6 (CMIP6) (Eyring et al., 2016) (Table S1 in Supporting Information S1). The historical simulations cover the period 1850–2014 and all forcings were included. The future projections for the period 2015–2100 are from the Scenario Model Intercomparison Project (ScenarioMIP) for CMIP6 (O'Neill et al., 2016), in which future greenhouse gas emission scenarios are derived from the Shared Socioeconomic Pathways (SSPs) with different climate policies. Projections from the four SSPs in Tier 1 of the ScenarioMIP for CMIP6 are adopted in the current work, namely SSP5-8.5, SSP3-7.0, SSP2-4.5, and SSP1-2.6. The four SSPs cover a major range of emission scenarios, from sustainability to fossil-fueled development. The time series of sea-level changes and the three selected variables are from the GCM grid closest to the study area.

2.6. GIS Analyses

Coastal geomorphology reconstruction based on eustatic sea level change and isostatic change for the latest 11 cal ka BP was analyzed using a shore displacement model from the Geological Survey of Sweden (SGU) (freely available at https://www.sgu.se/produkter-och-tjanster/geologiska-data/oppna-data/jordarter-oppna-data/ strandforskjutningsmodell/) based on calculations from Påsse and Andersson (2005) and Påsse and Daniels (2015). Data from the model were extracted for Gåsö and analyzed in ArcGIS pro (v. 2.9). From this data, bay exposure in terms of effective fetch (km) was calculated for the seagrass sites following the calculations outlined in Rogala (1997) by creating sight lines each 12° for a full circle (360°) with a maximum distance of the lines set to 3 km. The effective fetch was calculated for every 1,000 years over the past 11 cal ka BP. Based on the predicted SLR for the four different climate change scenarios (i.e., SSP5-8.5, SSP3-7.0, SSP2-4.5, and SSP1-2.6), the future sea levels and the resulting changes in shoreline were modeled in ArcGIS pro and calculated from digital elevation models (based on 0.1 altitude resolution LiDAR data from the Swedish National Land Survey) using the raster calculator function. Based on the predicted for the different climate change scenarios. The water depth curves of Gåsö were extracted from publicly available nautical charts obtained from the Swedish Maritime Administration.

2.7. Statistical Analyses

All statistical analyses were performed in R (v. 4.2.2). Principal component analysis (PCA) was used to explore common patterns in the different biogeochemical properties (i.e., OC, C/N, δ^{13} C, δ^{15} N, Br, Ca, Cl, Fe, K, Mn, Rb, Sr, Ti, Zr, MS, ACL, mean grain size and sediment sorting) analyzed along the cores. Prior to the PCA, the missMDA-package was used to impute missing values in the sediment profiles as not all biogeochemical analyses were performed in each sediment layer. We assessed the uncertainty associated with the data included in the PCA through multiple imputations with the function MIPCA. The results showed that the over-imputed values were acceptable and the PCA results were worth interpreting (Figure S1 in Supporting Information S1). The first two principal components (PC1 and PC2) explained a high proportion of the variance but showed a "horseshoe effect" that is an artifact typical of PCA when the PC-curves correlate to each other (Goodall, 1954), which was addressed by calculating the arc-length distance of PC1 and PC2. Finally, change point modeling (CPM) was used to identify change points along the arc-length distance profiles using the Beast-package (Zhao et al., 2019).







3. Results

3.1. Sediment Age Models

The mean (\pm SD) calibrated radiocarbon age at the bottom of the cores were 14.6 \pm 0.2 (188 cm) and 12.3 \pm 0.1 (165 cm) cal ka BP for Gåsö S and N, respectively. The mean (min-max) MAR was 0.0101 g cm⁻² year⁻¹ (0.0094–0.0107) and SAR was 0.0151 cm year⁻¹ (0.0140–0.0159) for Gåsö S and MAR was 0.0060 g cm⁻² year⁻¹ (0.0057–0.0062) and SAR was 0.0114 cm year⁻¹ (0.0108–0.0118) for Gåsö N (Figure S2 and Table S2 in Supporting Information S1). For Gåsö S, a large stone was encountered in the sediment core at a cumulative mass of around 24 g cm⁻² (corresponding to 70 cm sediment depth; Figure 3) and this section was therefore excluded in the Bacon model. In Gåsö N, one ¹⁴C date was omitted in the age model due to inversed age (Table S2 in Supporting Information S1).

3.2. Stratigraphy and Lithological Units

The bottom lithological units in both Gåsö S and N cores were composed of homogeneous brown clay (from 195 to 140 cm in Gåsö S and from 173 to 141 cm in Gåsö N) (Figure 2). In Gåsö S, brownish clay with shell fragments and pebbles was found from 140 to 104 cm, whereas homogeneous gray clay (without shell fragments) was found in Gåsö N from 141 to 106 cm. A layer of gray clay with shell fragments was found from 106 to 69 cm, which was also present in Gåsö S (104–85 cm). A 7–12 cm layer of gravel mixed with shells and clay was found between 72 and 60 cm in Gåsö S and between 69 and 60 cm in Gåsö N. The uppermost section in both cores (from ~60 cm to the surface), corresponding to the latest ~2.0 cal ka BP, contained organic-rich and dark brown sediment with seagrass rhizomes (Figure 2).





Figure 3. Summary of the physical and biogeochemical properties of the sediment, including organic carbon (OC), nitrogen (N), stable carbon isotopes (δ^{13} C), proportion of carbonate content (CaCO₃), dry bulk density (DBD), proportion of silt-clay (mud) content, degree of sorting (sorting coefficient), magnetic susceptibility (MS), average chain length (ACL) of *n*-alkanes (for homologs between C₁₇ and C₃₅) and seagrass biomass associated *n*-alkanes (i.e., C₁₇, C₁₉, and C₂₁) (a) along core depths and ages (shown as calibrated thousand years before present), and (b) zoomed in on the transition period of seagrass colonization (upper ~80 cm). Note that panel (b) shows sediment depth on the primary axis for better representation of the last ~2 ka cal BP. The black line in panels (a) and (b) indicates the first appearance of seagrass-derived sediment in the paleorecord, as evident by the introduction of *Zostera marina* biomass associated *n*-alkanes (Figure S3 in Supporting Information S1).

3.3. Physical and Biogeochemical Sediment Properties

The oldest sediment layers (14–8 cal ka BP) had relatively high MS (80–100 χ cm⁻³ g⁻¹) associated with lithogenic material and δ^{13} C values below –20‰ indicating terrestrial origin, which transitioned toward more marine signatures in shallower sediment layers (Figure 3a). At the bottom of the cores, the sediment was finer grained and well sorted between 14 and 8 cal ka BP and shifted toward coarser and more poorly sorted after 8 cal ka BP (Figure 3a), likely reflecting changes in the source material and deposition of finer grain-sized material (Figure 3). In the mid-section of the sediment profiles, the sediment was to a large degree composed of biogenic carbonate shells, and the highest carbonate content was found between 8 and 4.5 cal ka BP (Figure 3





Figure 4. Changes in sediment characteristics shown as biplot of PC1 and PC2 scores in relation to sediment depth indicated by colors (a), and arc-length distance with age (calibrated thousand years before present) (b). The horizontal orange (Gåsö S) and green (Gåsö N) lines in panel (b) show the change point probability for the arc-length distance of PC1 and PC2 for each core. The top panel in (a) shows the different shapes of the PC curves for the two cores, which is due to differences in sediment composition as seen in the PCA biplot (lower plot in a). The PCA biplot presents the relationship of sediment characteristics for PC1 and PC2, with colors indicating core depth. The arc-length distance in panel (b) represents the distance of PC1 and PC2 to the modelled curve, and was applied due to correlation of the two PC-axes. The black line in panel (b) indicates the first appearance of seagrass-derived sediment in the paleorecord. Section 1 shows the time frame prior to seagrass establishment (as a baseline), whereas Section 2 presents the period around seagrass colonization and Section 3 presents the seagrass stabilization phase (see Table 2).

and 4a). For the last 2 cal ka BP (from 60 to 62 cm to the surface), there was a change in the biogeochemical profiles with higher OC, N and silt-clay content, and lower DBD (Figure 3b), which was noticeable during the last \sim 1 cal ka BP (Figure 4b). From \sim 2 cal ka BP until present, the *n*-alkane depth profiles showed a higher contribution of C₁₇, C₁₉ and C₂₁ homologs, which are characteristic of *Z. marina* (Figure S3 in Supporting Information S1). The introduction of marine macrophytes was also reflected in an initial increase in average chain length, ACL, of the *n*-alkanes (Figure 3b).

3.4. Principal Component Analysis and Change Point Modeling

Two principal components explained 58% of the variability for which PC1 accounted for 36% and PC2 for 22% of the variance (Figure 4a). PC1 showed positive loadings (>0.6) for the XRF-elements Rb, Fe, K, Mn and Ti, and negative loadings (<-0.6) for δ^{13} C, OC, Cl and Br, whereas PC2 showed positive loadings (>0.6) for OC, Br and Zr, and negative loadings (<-0.6) for CN, Sr and Ca (Table 1). The relationship of different sediment properties varied along the cores with strong negative correlations between sediment depth and OC, δ^{13} C and Br, respectively, while elements such as Fe, Ti and MS were positively correlated to sediment depth (Figure 4a). CPM applied to the arc-length distance profiles for the two sites identified three change points in the core from Gåsö S at ~0.7 (48 cm), ~2.7 (68 cm) and ~10.2 cal ka BP (164 cm). At Gåsö N, five change points were suggested at ~1.1 (52 cm), ~2.6

Table 1

Factor Loadings of the Principal Component Analyses (PCA) Based on the Biogeochemical Data From Both Sediment Cores (Gåsö S and N) Combined

0		,
Biogeochemical variables	PC1	PC2
$\delta^{13}C$	-0.93	0.21
Cl	-0.82	0.36
Br	-0.64	0.70
OC	-0.55	0.73
Zr	-0.35	0.82
C/N	-0.14	-0.85
Sr	0.01	-0.57
$\delta^{15}N$	0.12	-0.09
ACL	0.12	-0.32
Mean grain size	0.17	0.33
Sediment sorting	0.18	-0.32
Ca	0.43	-0.62
MS	0.46	-0.01
Rb	0.68	-0.16
Fe	0.81	0.33
Κ	0.90	-0.29
Mn	0.93	-0.26
Ti	0.95	0.03

Note. Bold numbers show factor loadings of >0.6 for positive loadings and <-0.6 for negative loadings for PC1 and PC2. The data included in the PCA were organic carbon (OC), C:N-ratio (CN) and stable isotopes of C (δ^{13} C) and N (δ^{15} N), XRF-elements (Cl, Br, Zr, Sr, Ca, Rb, Fe, K, Mn and Ti), *n*-alkanes (average chain length, ACL), mean sediment grain-size distribution and sorting, and magnetic susceptibility (MS).

(68 cm), ~8.6 (128 cm), ~9.6 (134 cm) and ~11.5 cal ka BP (146 cm) (Figure 4b). Based on identified change points of the arc-length distance of PC1 and PC2, three main sections were distinguished that reflect the time frame of main interest for seagrass colonization (Figure 4 and Table 2). Section 1 is the time period prior to seagrass colonization, Section 2 marks the period for seagrass colonization, and Section 3 is defined as the seagrass stabilization period. Section 3 had up to 6–fold higher SAR than Section 1, whereas MAR was about 30%–50% faster in Section 3 than in Section 1 (Table 2). The CAR and NAR were also up to 24–fold higher in Section 1 (CAR: 8.2 ± 1.2 and 5.5 ± 1.7 g OC m⁻² year⁻¹, NAR: 0.75 ± 0.11 and 0.50 ± 0.31 g N m⁻² year⁻¹, NAR: 0.05 ± 0.01 and 0.02 ± 0.04 g N m⁻² year⁻¹).

3.5. Coastal Geomorphology and Isostatic Changes During Holocene

Based on the GIS (Geographic Information Systems) analysis, the effective fetch (km) for the Gåsö sites decreased during the latest 6 cal ka BP due to the isostatic uplift (Figure 5a), and in particular over the latest 2 cal ka BP when Gåsö was above the sea level creating a relatively sheltered environment (with 58%–72% lower effective fetch compared to the conditions prevalent about 7–11 cal ka BP). Over the last millennium, the effective fetch and shore-level have stabilized, and Gåsö has a similar coastal geomorphology as today (Figure 5b).

3.6. Modeling the Impacts of Climate Change to 2100

The climate models predict a sea-level increase ranging from 0.20 to 0.53 m for the different SSP scenarios by 2100, which could result in the immersion of the center of the Gåsö Island where postglacial sands are deposited (Figures 6a and 6b). In the high emission scenarios (SSP3-7.0 and SSP5-8.5), the SLR will create an opening in the middle of the island by 2100 and potentially change the hydrodynamics of the southern embayment. The predictions revealed a maximum of 3% increase in effective fetch by 2100 due to SLR (based on SSP5-8.5). The SST is predicted to increase in all SSP scenarios by $1.3-4.5^{\circ}$ C (mean over the year) in 2100 compared to the present SST

Table 2

Sediment Layer Thickness, Age, Sediment Accretion Rates and Organic Carbon and N Accumulation Rates for Three Sediment Sections Identified Based on Change Point Modeling Analyses of Arc-Length Distance for PC1 and PC2 in Gåsö S and N Cores

	Section 1 (Pri establis	Section 1 (Prior to seagrass establishment)		Section 2 (Seagrass colonization)		Section 3 (Seagrass stabilization)	
	Gåsö S	Gåsö N	Gåsö S	Gåsö N	Gåsö S	Gåsö N	
Sed. layer thickness (cm)	120	100	20	14	48	52	
Age (cal ka BP)	12.1 ± 0.2	14.3 ± 0.3	2.7 ± 0.4	2.6 ± 0.4	0.7 ± 0.3	1.1 ± 0.3	
SAR (cm year ⁻¹)	0.012 ± 0.008	0.008 ± 0.001	0.010 ± 0.003	0.009 ± 0.002	0.079 ± 0.021	0.058 ± 0.019	
MAR (g cm ^{-2} year ^{-1})	0.010 ± 0.002	0.006 ± 0.001	0.008 ± 0.001	0.005 ± 0.001	0.014 ± 0.002	0.009 ± 0.003	
%OC (weighted mean)	0.49	0.78	1.09	1.43	5.94	6.21	
%N (weighted mean)	0.05	0.04	0.09	0.12	0.55	0.56	
CAR (gOC $m^{-2} year^{-1}$)	0.51 ± 0.07	0.28 ± 0.05	0.51 ± 0.09	0.87 ± 0.12	8.18 ± 1.24	5.51 ± 1.68	
NAR (gN m ^{-2} year ^{-1})	0.05 ± 0.01	0.02 ± 0.04	0.04 ± 0.01	0.06 ± 0.01	0.75 ± 0.11	0.50 ± 0.31	

Note. Section 1 shows the period before establishment of the seagrass (as a baseline), whereas Section 2 shows the time around seagrass colonization and Section 3 presents the period for seagrass stabilization. Cal ka BP, calibrated thousand years before present; SAR, sediment accumulation rate; MAR, mass accumulation rate; CAR, organic carbon accumulation rate; NAR, nitrogen accumulation rate. All accumulation rates are presented as mean \pm SD.





Figure 5. Changes in effective fetch (km) (a), and land uplift for 1, 2, 4, and 11 cal ka BP (calibrated thousand years before present) (b). Modeled data on shore displacement were downloaded from the Swedish Geological Survey and show the isostatic uplift and shore-level changes at Gåsö. The black line in panel (a) shows the first appearance of seagrass-derived sediment in the paleorecord.









Figure 7. Predicted annual mean sea surface temperature (SST) (a), salinity (b), and surface runoff (c) changes until 2100 for four Shared Socioeconomic Pathway (SSP) scenarios. Data are ensemble means from CMIP6 global climate models (GCMs), retrieved for the closest grid point to Gåsö (Eyring et al., 2016; O'Neill et al., 2016). For information on GCMs included in the historical and scenario ensembles, see Table S1 in Supporting Information S1.

(calculated as an average of the SST for 2004–2014) (Figure 7a). The salinity will decrease in all SSP scenarios from around 27–28 at present down to 25–26 (Figure 7b). However, large variation and no clear trends were seen in the regional models for surface runoff (Figure 7c).

4. Discussion

The reconstruction of the past ~14–12 cal ka BP revealed that the colonization of eelgrass around 2,000 years ago transformed the rate and shape of the biogeochemical sink through ecosystem stabilization and sediment accretion, leading to higher concentrations of finer (mud) grain size particles, and an increase in accumulation rates of up to 24–fold for sedimentary OC and N. The onset of seagrass establishment was likely related to the change in coastal geomorphology during this time period, resulting in decreased water depth and hydrodynamic exposure of Gåsö, which created a favorable environment for seagrass growth. After the initial colonization period, the most drastic biogeochemical changes occurred between 1.1 and 0.7 cal ka BP. This time period, corresponding to the MCA, was characterized by warmer temperatures and lower storm activity (Hass, 1996), and these changes in climate could also have further stimulated seagrass productivity and meadow stability, which in turn enhanced the accumulation of organic-rich sediment even further. This underscores the significance of both abiotic and biotic factors in facilitating seagrass establishment and sustaining the essential ecosystem services provided by healthy seagrass meadows.

4.1. Environmental Conditions Conductive to Seagrass Colonization

The deglaciation of Northern Europe during the late Pleistocene and early Holocene resulted in an isostatic uplift (Rosentau et al., 2021; Stroeven et al., 2016) that led to coastal geomorphology changes at Gåsö. Sediment derived from the melting ice sheet and outflow of freshwater and associated clay-rich sediments from the Baltic basin through south-central Sweden deposited in coastal settings (Bergsten & Dennegård, 1988; Gyllencreutz, 2005; Gyllencreutz & Kissel, 2006), which was observed in the deepest layers of the Gåsö cores composed of brown clay of terrestrial lithogenic material rich in mud, as indicated by elemental composition, δ^{13} C isotopic signatures below -20% and increased MS values. The freshwater outflow decreased around 10.7 cal ka BP (Gyllencreutz & Kissel, 2006) and the opening of the English Channel at about 8.5 cal ka BP (Conradsen & Heier-Nielsen, 1995) resulted in higher hydrodynamic activity and a larger inflow of marine waters from the Atlantic that carried coarser sediments into Skagerrak (Gyllencreutz, 2005). This coarsening of sediment was also clearly visible in the cores from Gåsö at that time, with more poorly sorted sediment reflecting a higher hydrodynamic activity. There were also two change points identified in Gåsö N at ~9.7 and ~8.6 cal ka BP, which might reflect this change in oceanic circulation with the inflow of Atlantic water. Gåsö N also had a more pronounced shift in MS values compared to Gåsö S, indicating a change from terrestrial to marine influence (López-Merino et al., 2017).

A general gradual increase in the marine influence from 14 to 12 cal ka BP until present was marked by shifts in both cores around 2.7-2.6 cal ka BP (Section 2) that resulted in enhanced sediment, OC and N accretion as well as higher carbonate content likely due to the isostatic uplift with the shore-level becoming similar to present conditions that created a more sheltered and stable environment. Seagrass colonization was detected around 2 cal ka BP based on shifts in δ^{13} C, *n*-alkanes (with the appearance of the seagrass-related C₁₇, C₁₉, and C₂₁ *n*alkanes; Chevalier et al., 2015; Rosenbauer et al., 2006; which were absent prior to this time period), silt-clay content, elemental composition (including higher Br and Cl associated to the increased organic matter content) and sediment density, which shows alteration of the sediment characteristics similar to pedogenesis in terrestrial soils (Piñeiro-Juncal et al., 2020). The transformation of the sediment was likely triggered by the higher depositional rate linked to the reduction in hydrodynamic exposure. These levels of hydrodynamic exposure are prevalent within the range of contemporary Zostera spp. distribution (Dahl et al., 2020; Prentice et al., 2019; Short et al., 2007). The colonization period of seagrass also coincided with the RWP, which was characterized by regionally warm temperatures (Neukom et al., 2019). Zostera marina can tolerate a large range of temperatures with an upper limit of approximately 25-30°C (Nejrup & Pedersen, 2008), and while the optimum water temperature for plant growth is likely site-specific, it has been shown that productivity of Z. marina increases with temperature (within the upper thermal limit) (Rasmusson et al., 2019), which could have stimulated the spread of seagrasses. During the time of seagrass establishment, the oceanographic conditions were more dominated by Atlantic waters and less turbidity (Erbs-Hansen et al., 2012). After the colonization phase (Section 2), a shift in the CPM was identified for both cores due to the higher presence of marine-derived organic matter and more wellsorted sediments of finer grain size composition (Section 3). This change in the biogeochemical trends with enhanced OC and N accumulation occurred at 1.1-0.7 cal ka BP, which coincides with the MCA, a period of higher temperature (Mann et al., 2009) that may have favored seagrass productivity and growth even further. The last millennium constitutes a stable period suitable for seagrass growth, which is suggested by the more consistent hydrodynamic exposure and shore-level at Gåsö.

4.2. Ecosystem Services Related to Seagrass Colonization

The shifts seen in the sediment record following eelgrass establishment and stabilization phases consisted of enhanced sediment and mud accretion, and organic matter accumulation that could be related to both changes in the coastal geomorphology (a shallower and more sheltered environment) and through enhanced seagrass biomass and productivity across the region. Indeed, the effect of the seagrass canopy likely reduced water turbidity, resulting in an overall increase in ecosystem stability and productivity, and associated increases in the accumulation of OC and N. The establishment of the seagrass was a crucial requirement for the modification of sediment biogeochemical properties, with approximately 47%–48% of the organic matter originating from the seagrass biomass itself, while most of the remainder (43%–45%) was of macroalgae origin (Dahl et al., 2023). However, the carbon- (CAR: 6–8 g OC m⁻² year⁻¹) and nitrogen accumulation rates (NAR: 0.5–0.8 g N m⁻² year⁻¹) at Gåsö for the latest ~0.7 to 1 cal ka BP (Section 3) are still lower than accumulation rates in *Zostera* spp. meadows in general (Martins et al., 2022; Prentice et al., 2020) and in the lower range compared to levels found in other studies in the Skagerrak-Kattegat region, which estimated CAR between 6 and

134 g OC m⁻² year⁻¹ and NAR from 0.7 to 14 g N m⁻² year⁻¹ (Dahl et al., 2023; Leiva-Dueñas et al., 2023). However, these studies only assessed accumulation over shorter time periods (the last decades to century), while long-term accumulation rates tend to be lower for seagrass in general due to diagenesis and remineralization of the organic matter over time (Belshe et al., 2019). This indicates that while the organic matter has been gradually accumulating since the onset of seagrass colonization (Section 2), it has done so at a comparatively faster rate (20to 24-fold) since seagrass stabilization (Section 3 in comparison to Section 1). Although sediments tend to have a lower density at the surface due to condensation with depth, the clear decrease in DBD and increased sorting of finer sediment components following the onset of the organic-rich seagrass sediment build-up after the seagrass establishment shows the effect of the sediment stabilization by the seagrass structure.

4.3. Climate Change Impacts on Seagrass Plant Health Status

The predicted SLR by 2100 alone will likely not impact the present seagrass distribution at Gåsö. In fact, with an increase of about 0.5 m (for SSP5-8.5 in 2100), the seagrass meadow in the southern bay (Gåsö S) has the potential to colonize an extended area, primarily the inner part of the bay that is currently too shallow (less than 0.4 m) for Z. marina. The lateral expansion of the rhizomes from an established Z. marina meadow has been estimated at 16 cm year⁻¹ (Olesen & Sand-Jensen, 1994), which would allow the seagrass meadow to adjust to the predicted SLR and likely thrive along the deeper part of the bay based on the current depth range of Z. marina on the Swedish Skagerrak coast. In contrast, the GIS-modeled SLR leads to a submerging of the middle part of the island, creating a channel connecting the southern and northern bays that will likely remobilize postglacial sand that could result in smothering of the seagrass meadow. This center part of the island surfaced around 800 to 700 years ago due to the isostatic uplift, which occurred during the time period of seagrass stabilization for Gåsö S (Section 3), and an opening of the channel in the future scenarios may change the hydrodynamics of Gåsö and increase the hydrodynamic activity. Though, SLR alone would not affect the hydrodynamic exposure with a calculated maximum increase of 3% in effective fetch by 2100 for the highest climate scenario (SSP5-8.5), increased sediment transport in a north-south direction due to the opening of the channel could lead to increased turbidity locally, which might negatively affect the contemporary distribution of seagrass at Gåsö. However, the potential impact from turbidity and sedimentation might be counteracted by increasing the available area for seagrass expansion through the deepening of the southern bay.

The projected regional SST increase until 2100 (with a predicted annual mean ranging from 11 to 14° C and mean summer maximum temperatures of $18-21^{\circ}$ C in 2100 for the different climate scenarios) is within the growth range for *Z. marina*, which generally has a high tolerance for increased temperatures (Rasmusson et al., 2020). Historical periods of increased sea water temperature (i.e., the RWP and the MCA) were still lower compared to today (Lapointe et al., 2020) and therefore present temperatures are likely representing the warmest period during the colonization history of the seagrass meadows at Gåsö. Nejrup and Pedersen (2008) identified the optimum water temperature for *Z. marina* to range from 10 to 20° C, with an increase in mortality and a decrease in photosynthetic activity and growth occurring at $25-30^{\circ}$ C. However, the SST data do not account for extreme events such as heat waves, which are expected to increase in the future. Such heat wave events could substantially increase the water temperature for short periods of time and result in major seagrass loss and negatively affect ecosystem services (Aoki et al., 2020, 2021). Although an increase in water temperature, either as a short-term single event or as a gradual increase over time, might not strongly affect seagrass productivity and growth in the Skagerrak region, a higher temperature can lead to reduced water quality through stimulation of filamentous-or microalgae growth (Moore et al., 2012), which likely has a stronger impact on the seagrass health status than the increased SST alone.

This decrease in modeled salinity (reaching around 25 for SSP3-7.0 and SSP5-8.5 in 2100) is also within the tolerance limit for *Z. marina*, which shows large plasticity in response to salinity as the species can grow in brackish environments down to around 5–6 (Boström et al., 2014). However, the salinity tolerance is also population-specific with *Z. marina* growing in the marine environment (as the seagrass at Gåsö) being less adapted to low salinity (Salo et al., 2014). The projected salinity decrease is still within the tolerance limit and will likely not affect the seagrass plant health to a large extent (Hellblom & Björk, 1999; Zhang et al., 2022). Reconstruction of salinity and sea water temperature in the Skagerrak region (Jiang et al., 1998; Ni et al., 2021) shows that these environmental parameters have likely remained within the tolerance limit for *Z. marina* growth throughout the Holocene.

19449224, 2024, 3, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2023GB008039 by Goteborgs, Wiley Online Library on [29/03/2024]. See the Terms

and Condition

° on

Wiley Online Library for rules

of use; OA articles are governed by the applicable Creative Commons Licens

4.4. Opportunities for Seagrass Conservation and Restoration

Although the environmental conditions thousands of years ago were in many ways different, reconstructing the interactions between environmental conditions and the colonization and establishment by seagrasses together with the prediction of potential climate change impacts provides important information for seagrass conservation and restoration, which could help guide management. This provides further evidence that the sediment characteristics and somewhat sheltered conditions (leading to reduced turbidity) are of importance for successful Z. marina replantation and re-colonization, as shown by several restoration projects (Orth et al., 2010). Positive feedback likely occurred following the establishment of seagrasses through enhanced water quality and irradiance by reducing resuspension and increasing the deposition of suspended particles, which resulted in enhanced clay and silt, and OC and N accumulation. Moksnes et al. (2021) proposed that suitable restoration sites in the Skagerrak-Kattegat area should have mud contents of less than 34% and OC levels below 5%. If the mud content and resuspension of sediment are high and the seagrass meadow is in an early stage of establishment, likely the seagrass plants cannot stabilize the sediment enough for sufficient irradiance to support plant growth and reproduction (Moksnes et al., 2016) and the Z. marina meadows at Gåsö established in coarse sediment with low mud ($\sim 5-6\%$) and OC contents (0.5%-1.5%). This finding aligns with the ongoing replantation projects on the Swedish Skagerrak coast, where several tons of sand have been added to the bottom within a km-wide project area in order to stabilize the sediment prior to replantation of Z. marina shoots.

This study shows critical coastal geomorphology requirements for the success of seagrass establishment and colonization, including a low hydrodynamic exposure (with an effective fetch of <1.3 Lf [km]) and overall stability of the environment. This confirms the findings of previous studies showing that site selection for restoration projects needs to be carefully evaluated for a successful outcome of the restoration effort (Short et al., 2002; van Katwijk et al., 2009). As long-term monitoring of coastal habitats is commonly lacking, paleoreconstruction could help identify areas that are environmentally stable, which could be more suitable for habitat conservation, and to assess carbon and nitrogen abatement following seagrass establishment (i.e., estimated at between 63 and 68 Mg OC ha⁻¹; Dahl et al., 2023; and 5 and 7 Mg N ha⁻¹ over the past 2 cal ka BP) related to for example, Nationally Determined Contributions. The historical paleorecord in combination with regional climate models is useful in the assessment of risk related to climate change for cold-temperate Z. marina and provides insights into potential impacts linked to climate change scenarios (e.g., SLR, precipitation and storminess), such as the enhanced build-up of sedimentary organic matter during relatively warmer periods of increased temperature in the late Holocene. Zostera marina in Northern Europe inhabits the lower end of its thermal tolerance range and therefore, it has the potential to thrive under predicted warming scenarios (Rasmusson et al., 2019). However, the Z. marina meadows in Skagerrak are subjected to multiple threats (both from single events and through cumulative impacts; Baden et al., 2003; Moksnes et al., 2008, 2018) that might weaken their resilience against climate change threats (Björk et al., 2008; Unsworth et al., 2015). This multiplicity of potential disturbances needs to be considered to ensure successful habitat conservation.

Successful restoration and re-vegetation of seagrass meadows has been shown to increase ecosystem services derived from the build-up of sediment deposits (Greiner et al., 2013; Marbà et al., 2015) and enhance biodiversity, among other benefits for the people and the planet (Orth et al., 2020). However, many restoration projects have failed owing to the environmental conditions not being suitable for seagrass growth (van Katwijk et al., 2016). This study provides further evidence of the importance of environmental conditions for natural seagrass colonization and this knowledge could thus help to identify sites suitable for successful seagrass conservation and restoration efforts.

Data Availability Statement

Data set for the biogeochemical analysis in this article can be found in Dahl et al. (2024), available at https:// zenodo.org/records/10559287.

References

Andrén, T., Björck, S., Andrén, E., Conley, D., Zillén, L., & Anjar, J. (2011). The development of the Baltic Sea basin during the last 130 ka (pp. 75-97). https://doi.org/10.1007/978-3-642-17220-5_4

Aoki, L. R., McGlathery, K. J., Wiberg, P. L., & Al-Haj, A. (2020). Depth affects seagrass restoration success and resilience to marine heat wave disturbance. Estuaries and Coasts, 43(2), 316-328. https://doi.org/10.1007/s12237-019-00685-0

Carina Johansson for their assistance during the laboratory work. We also thank the Kristineberg Center for Marine Research and their scientific diving facility. The authors would like to thank Beat Gasser and Inés Sanz-Álvarez for contributing to the ²¹⁰Pb analysis. We are also thankful to the two anonymous reviewers for their comments and feedback. The IAEA is grateful for the support provided by its Marine Environment Laboratories through the Government of the Principality of Monaco. Funding was provided by the foundation for Baltic and East European studies (Östersjöstiftelsen) (Grant 21-GP-0005 and 21-PD2-0002), FORMAS (Grant 2021-01280), Helge Ax:son Johnson foundation (Grant F21-0103), Bolin Centre for climate research, and Albert och Maria Bergström foundation. O.S. was supported by I + D + i projects RYC2019-027073-I and PIE HOLOCENO 20213AT014 funded by MCIN/AEI/10.13039/ 501100011033 and FEDER.

- Aoki, L. R., McGlathery, K. J., Wiberg, P. L., Oreska, M. P. J., Berger, A. C., Berg, P., & Orth, R. J. (2021). Seagrass recovery following marine heat wave influences sediment carbon stocks. *Frontiers in Marine Science*, 7, 576784. https://doi.org/10.3389/fmars.2020.576784
- Arias-Ortiz, A., Masqué, P., Glass, L., Benson, L., Kennedy, H., Duarte, C. M., et al. (2020). Losses of soil organic carbon with deforestation in mangroves of Madagascar. *Ecosystems*, 24(1), 1–19. https://doi.org/10.1007/s10021-020-00500-z
- Asplund, M. E., Dahl, M., Ismail, R. O., Arias-Ortiz, A., Deyanova, D., Franco, J. N., et al. (2021). Dynamics and fate of blue carbon in a mangrove–seagrass seascape: Influence of landscape configuration and land-use change. *Landscape Ecology*, 8(5), 1489–1509. https://doi.org/ 10.1007/s10980-021-01216-8
- Baden, S., Gullström, M., Lundén, B., Pihl, L., & Rosenberg, R. (2003). Vanishing seagrass (Zostera marina, L.) in Swedish coastal waters. Ambio, 32(5), 374–377. https://doi.org/10.1579/0044-7447-32.5.374
- Barbier, E. B., Hacker, S. D., Kennedy, C., Koch, E. W., Stier, A. C., & Silliman, B. R. (2011). The value of estuarine and coastal ecosystem services. *Ecological Monographs*, 81(2), 169–193. https://doi.org/10.1890/10-1510.1
- Belshe, E. F., Sanjuan, J., Leiva-Dueñas, C., Piñeiro-Juncal, N., Serrano, O., Lavery, P., & Mateo, M. A. (2019). Modeling organic carbon accumulation rates and residence times in coastal vegetated ecosystems. *Journal of Geophysical Research: Biogeosciences*, 124(11), 3652–3671. https://doi.org/10.1029/2019JG005233
- Bengtsson, L., & Enell, M. (1986). Chemical analysis. In B. E. Berglund (Ed.), Handbook of holocene palaeoecology and palaeohydrology. Wiley & Sons.
- Bergsten, H., & Dennegård, B. (1988). Late Weichselian-Holocene foraminiferal stratigraphy and palaeohydrogrphic changes in the Gothenburg area, southwestern Sweden. *Boreas*, 17(2), 229–242. https://doi.org/10.1111/j.1502-3885.1988.tb00553.x
- Berner, K. S., Koç, N., Godtliebsen, F., & Divine, D. (2011). Holocene climate variability of the Norwegian Atlantic Current during high and low solar insolation forcing. *Paleoceanography*, 26(2), PA2220. https://doi.org/10.1029/2010pa002002
- Björck, S. (2008). The Late Quaternary development of the Baltic Sea basin. In A. Team (Ed.), Assessment of climate change for the Baltic Sea Basin (pp. 398–407). Springer-Verlag Berlin Heidelberg. https://doi.org/10.1139/t91-063
- Björk, M., Short, F., Mcleod, E., & Beer, S. (2008). Managing seagrasses for resilience to climate change. IUCN.
- Blaauw, M., & Christeny, J. A. (2011). Flexible paleoclimate age-depth models using an autoregressive gamma process. Bayesian Analysis, 6(3), 457–474. https://doi.org/10.1214/11-BA618
- Blott, S. J., & Pye, K. (2001). Gradistat: A grain size distribution and statistics package for the analysis of unconsolidated sediments. Earth Surface Processes and Landforms, 26(11), 1237–1248. https://doi.org/10.1002/esp.261
- Boström, C., Baden, S., Bockelmann, A.-C., Dromph, K., Fredriksen, S., Gustafsson, C., et al. (2014). Distribution, structure and function of Nordic eelgrass (*Zostera marina*) ecosystems: Implications for coastal management and conservation. *Aquatic Conservation: Marine and Freshwater Ecosystems*, 24(3), 410–434. https://doi.org/10.1002/aqc.2424
- Cheng, L., Trenberth, K. E., Gruber, N., Abraham, J. P., Fasullo, J. T., Li, G., et al. (2020). Improved estimates of changes in upper ocean salinity and the hydrological cycle. *Journal of Climate*, 33(23), 10357–10381. https://doi.org/10.1175/jcli-d-20-0366.1
- Chevalier, N., Savoye, N., Dubois, S., Lama, M. L., David, V., Lecroart, P., et al. (2015). Precise indices based on n-alkane distribution for quantifying sources of sedimentary organic matter in coastal systems. Organic Geochemistry, 88, 69–77. https://doi.org/10.1016/j. orggeochem.2015.07.006
- Conradsen, K., & Heier-Nielsen, S. (1995). Holocene paleoceanography and paleoenvironments of the Skagerrak-Kattegat, Scandinavia. Paleoceanography, 10(4), 801–813. https://doi.org/10.1029/95pa01142
- Cullen-Unsworth, L., & Unsworth, R. (2013). Seagrass meadows, ecosystem services, and sustainability. Environment: Science and Policy for Sustainable Development, 55(3), 14–28. https://doi.org/10.1080/00139157.2013.785864
- Dahl, M., Asplund, M. E., Bergman, S., Björk, M., Braun, S., Löfgren, E., et al. (2023). First assessment of seagrass carbon accumulation rates in Sweden: A field study from a fjord system at the Skagerrak coast. PLoS Climate, 2(1), e0000099. https://doi.org/10.1371/journal.pclm.0000099
- Dahl, M., Asplund, M. E., Björk, M., Deyanova, D., Infantes, E., Isaeus, M., et al. (2020). The influence of hydrodynamic exposure on carbon storage and nutrient retention in eelgrass (*Zostera marina* L.) meadows on the Swedish Skagerrak coast. *Scientific Reports*, 10, 1–13. https:// doi.org/10.1038/s41598-020-70403-5
- Dahl, M., Deyanova, D., Gütschow, S., Asplund, M. E., Lyimo, L. D., Karamfilov, V., et al. (2016). Sediment properties as important predictors of carbon storage in *Zostera marina* meadows: A comparison of four European areas. *PLoS One*, 11(12), e0167493. https://doi.org/10.1371/ journal.pone.0167493
- Dahl, M., Gullström, M., Bernabeu, I., Serrano, O., Leiva-Dueñas, C., Linderholm, H. W., et al. (2024). A 2,000-year record of eelgrass (Zostera marina L.) colonization shows substantial gains in blue carbon storage and nutrient retention [Dataset]. Zenodo. https://doi.org/10.5281/ zenodo.10559287
- de los Santos, C. B., Scott, A., Arias-Ortiz, A., Jones, B., Kennedy, H., Mazarrasa, I., et al. (2020). Seagrass ecosystem services: Assessment and scale of benefits. In *Out of the blue: The value of seagrasses to the environment and to people* (pp. 19–21).
- Erbs-Hansen, D. R., Knudsen, K. L., Gary, A. C., Gyllencreutz, R., & Jansen, E. (2012). Holocene climatic development in Skagerrak, eastern North Atlantic: Foraminiferal and stable isotopic evidence. *The Holocene*, 22(3), 301–312. https://doi.org/10.1177/0959683611423689
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Folk, R. R. L., & Ward, W. W. C. (1957). Brazos River Bar: A study in the significance of grain size parameters. SEPM Journal of Sedimentary Research, 27(1), 3–26. https://doi.org/10.1306/74D70646-2B21-11D7-8648000102C1865D
- Fox-Kemper, B., Hewitt, H. T., Xiao, C., Aðalgeirsdóttir, G., Drijfhout, S. S., Edwards, T. L., et al. (2021). Ocean, cryosphere and sea level change. In V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Eds.), *Climate change 2021: The physical science basis. Contribution of Working Group I to the sixth assessment report of the Intergovernmental Panel on Climate Change* (pp. 1211–1362). Cambridge University Press. https://doi.org/10.1017/9781009157896.011
- Garner, G. G., Hermans, T., Kopp, R. E., Slangen, A. B. A., Edwards, T. L., Levermann, A., et al. (2021). IPCC AR6 sea-level rise projections. Version 20210809. PO.DAAC. Retrieved from https://podaac.jpl.nasa.gov/announcements/2021-08-09-Sea-level-projections-from-the-IPCC-6th-Assessment-Report
- Goodall, D. W. (1954). Objective methods for the classification of vegetation. III. An essay in the use of factor analysis. *Australian Journal of Botany*, 2(3), 304–324. https://doi.org/10.1071/bt9540304
- Greiner, J. T., McGlathery, K. J., Gunnell, J., & McKee, B. A. (2013). Seagrass restoration enhances "blue carbon" sequestration in coastal waters. *PLoS One*, 8(8), e72469. https://doi.org/10.1371/journal.pone.0072469
- Gyllencreutz, R. (2005). Late Glacial and Holocene paleoceanography in the Skagerrak from high-resolution grain size records. *Palaeogeography, Palaeoclimatology, Palaeoecology, 222*(3–4), 344–369. https://doi.org/10.1016/j.palaeo.2005.03.025

- Gyllencreutz, R., Backman, J., Jakobsson, M., Kissel, C., & Arnold, E. (2006). Postglacial palaeoceanography in the Skagerrak. *The Holocene*, *16*(7), 975–985. https://doi.org/10.1177/0959683606h1988rp
- Gyllencreutz, R., & Kissel, C. (2006). Lateglacial and Holocene sediment sources and transport patterns in the Skagerrak interpreted from highresolution magnetic properties and grain size data. *Quaternary Science Reviews*, 25(11–12), 1247–1263. https://doi.org/10.1016/j.quascirev. 2005.11.002
- Håkansson, S. (1970). University of Lund radiocarbon dates III. Radiocarbon, 12(2), 534–552. https://doi.org/10.1017/s0033822200008250
- Håkansson, S. (1987). University of Lund radiocarbon dates XX. Radiocarbon, 29(3), 353–379. https://doi.org/10.1017/s0033822200043769
 Hass, H. C. (1996). Northern Europe climate variations during late Holocene: Evidence from marine Skagerrak. Palaeogeography, Palaeoclimatology, Palaeoecology, 123(1–4), 121–145. https://doi.org/10.1016/0031-0182(95)00114-x
- Hatfield, R. G., Stoner, J. S., Carlson, A. E., Reyes, A. V., & Housen, B. A. (2013). Source as a controlling factor on the quality and interpretation of sediment magnetic records from the northern North Atlantic. *Earth and Planetary Science Letters*, 368, 69–77. https://doi.org/10.1016/j.epsl. 2013.03.001
- Heaton, T. J., Köhler, P., Butzin, M., Bard, E., Reimer, R. W., Austin, W. E. N., et al. (2020). Marine20—The marine radiocarbon age calibration curve (0–55,000 cal BP). *Radiocarbon*, 62(4), 779–820. https://doi.org/10.1017/rdc.2020.68
- Heiri, O., Lotter, A. F., & Lemcke, G. (2001). Loss on ignition as a method for estimating organic and carbonate content in sediments: Reproducibility and comparability of results. *Journal of Paleolimnology*, 25(1), 101–110. https://doi.org/10.1023/a:1008119611481
- Hellblom, F., & Björk, M. (1999). Photosynthetic responses in *Zostera marina* to decreasing salinity, inorganic carbon content and osmolality. *Aquatic Botany*, 65(1–4), 97–104. https://doi.org/10.1016/s0304-3770(99)00034-0
- IPCC. (2021). Summary for policymakers. In V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, et al. (Eds.), Climate change 2021: The physical science basis. Contribution of Working Group I to the sixth assessment report of the Intergovernmental Panel on Climate Change. Jaffé, R., Mead, R., Hernandez, M. E., Peralba, M. C., & DiGuida, O. A. (2001). Origin and transport of sedimentary organic matter in two
- subtropical estuaries: A comparative, biomarker-based study. Organic Geochemistry, 32(4), 507–526. https://doi.org/10.1016/S0146-6380(00) 00192-3
- Jakobsson, M., Björck, S., Alm, G., Andrén, T., Lindeberg, G., & Svensson, N. O. (2007). Reconstructing the Younger Dryas ice dammed lake in the Baltic Basin: Bathymetry, area and volume. *Global and Planetary Change*, 57(3–4), 355–370. https://doi.org/10.1016/j.gloplacha.2007. 01.006
- Jiang, H., Björck, S., & Svensson, N. (1998). Reconstruction of Holocene sea-surface salinity in the Skagerrak–Kattegat: A climatic and environmental record of Scandinavia. Journal of Quaternary Science: Published for the Quaternary Research Association, 13(2), 107–114. https://doi.org/10.1002/(sici)1099-1417(199803/04)13:2<107::aid-jqs336>3.3.co;2-r
- Kennedy, H., Beggins, J., Duarte, C. M., Fourqurean, J. W., Holmer, M., Marbà, N., & Middelburg, J. J. (2010). Seagrass sediments as a global carbon sink: Isotopic constraints. *Global Biogeochemical Cycles*, 24(4), GB4026. https://doi.org/10.1029/2010GB003848
- Kindeberg, T., Röhr, E., Moksnes, P.-O., Boström, C., & Holmer, M. (2019). Variation of carbon contents in eelgrass (Zostera marina) sediments implied from depth profiles. *Biology Letters*, 15(6), 20180831. https://doi.org/10.1098/rsbl.2018.0831
- Krause-Jensen, D., Carstensen, J., Nielsen, S. L., Dalsgaard, T., Christensen, P. B., Fossing, H., & Rasmussen, M. B. (2011). Sea bottom characteristics affect depth limits of eelgrass Zostera marina. Marine Ecology Progress Series, 425, 91–102. https://doi.org/10.3354/ meps09026
- Lapointe, F., Bradley, R. S., Francus, P., Balascio, N. L., Abbott, M. B., Stoner, J. S., et al. (2020). Annually resolved Atlantic sea surface temperature variability over the past 2,900 y. *Proceedings of the National Academy of Sciences*, 117(44), 27171–27178. https://doi.org/10. 1073/pnas.2014166117
- Lei, J., Schaefer, R., Colarusso, P., Novak, A., Simpson, J. C., Masqué, P., & Nepf, H. (2023). Spatial heterogeneity in sediment and carbon accretion rates within a seagrass meadow correlated with the hydrodynamic intensity. *Science of the Total Environment*, 854, 158685. https:// doi.org/10.1016/j.scitotenv.2022.158685
- Leiva-Dueñas, C., Graversen, A. E. L., Banta, G. T., Holmer, M., Masque, P., Stæhr, P. A. U., & Krause-Jensen, D. (2023). Capturing of organic carbon and nitrogen in eelgrass sediments of southern Scandinavia. *Limnology and Oceanography*, 68(3), 631–648. https://doi.org/10.1002/ lno.12299
- Leiva-Dueñas, C., Martínez Cortizas, A., Piñeiro-Juncal, N., Díaz-Almela, E., Garcia-Orellana, J., & Mateo, M. A. (2021). Long-term dynamics of production in western Mediterranean seagrass meadows: Trade-offs and legacies of past disturbances. *Science of the Total Environment*, 754, 142117. https://doi.org/10.1016/j.scitotenv.2020.142117
- Lima, M. D. A. C., Ward, R. D., & Joyce, C. B. (2020). Environmental drivers of sediment carbon storage in temperate seagrass meadows. *Hydrobiologia*, 847(7), 1773–1792. https://doi.org/10.1007/s10750-019-04153-5
- Loo, L.-O., & Isaksson, I. (2015). Stora bestånd av ålgräs förlorade i Kattegatt. In M. Svärd, T. Johansen Lilja, & D. Hansson (Eds.), Havet 1888 (pp. 42–43). Havsmiljöinstitutet.
- López-Merino, L., Colás-Ruiz, N. R., Adame, M. F., Serrano, O., Martínez Cortizas, A., & Mateo, M. A. (2017). A six thousand-year record of climate and land-use change from Mediterranean seagrass mats. *Journal of Ecology*, 105(5), 1267–1278. https://doi.org/10.1111/1365-2745. 12741
- López-Sáez, J. A., López-Merino, L., Mateo, M. Á., Serrano, Ó., Pérez-Díaz, S., & Serrano, L. (2009). Palaeoecological potential of the marine organic deposits of *Posidonia oceanica*: A case study in the NE Iberian Peninsula. *Palaeogeography, Palaeoclimatology, Palaeoecology*, 271(3–4), 215–224. https://doi.org/10.1016/j.palaeo.2008.10.020
- Macreadie, P. I., Allen, K., Kelaher, B. P., Ralph, P. J., & Skilbeck, C. G. (2012). Paleoreconstruction of estuarine sediments reveal humaninduced weakening of coastal carbon sinks. *Global Change Biology*, 18(3), 891–901. https://doi.org/10.1111/j.1365-2486.2011.02582.x
- Macreadie, P. I., Atwood, T. B., Kelleway, J. J., Kennedy, H., Lovelock, C. E., & Serrano, O. (2021). Blue carbon as a natural climate solution. *Nature Reviews Earth and Environment*, 2, 0123456789. https://doi.org/10.1038/s43017-021-00224-1
- Mann, M. E., Zhang, Z., Rutherford, S., Bradley, R. S., Hughes, M. K., Shindell, D., et al. (2009). Global signatures and dynamical origins of the Little Ice Age and Medieval Climate Anomaly. *Science*, 326(5957), 1256–1260. https://doi.org/10.1126/science.1177303
- Marbà, N., Arias-Ortiz, A., Masqué, P., Kendrick, G. A., Mazarrasa, I., Bastyan, G. R., et al. (2015). Impact of seagrass loss and subsequent revegetation on carbon sequestration and stocks. *Journal of Ecology*, 103(2), 296–302. https://doi.org/10.1111/1365-2745.12370
- Marsh, J. A., Dennison, W. C., & Alberte, R. S. (1986). Effects of temperature on photosynthesis and respiration in eelgrass (Zostera marina L.). Journal of Experimental Marine Biology and Ecology, 101(3), 257–267. https://doi.org/10.1016/0022-0981(86)90267-4
- Martins, M., Carmen, B., Masqué, P., Carrasco, A. R., Veiga-Pires, C., & Santos, R. (2022). Carbon and nitrogen stocks and burial rates in intertidal vegetated habitats of a mesotidal coastal lagoon. *Ecosystems*, 25(2), 372–386.
- Masson-Delmotte, V., Zhai, P., Pirani, A., Connors, S. L., Péan, C., Berger, S., et al. (2021). Climate change 2021: The physical science basis. Contribution of Working Group I to the sixth assessment report of the Intergovernmental Panel on Climate Change (Vol. 2).

- Mateo, M. A., Renom, P., & Michener, R. H. (2010). Long-term stability in the production of a NW Mediterranean Posidonia oceanica (L.) Delile meadow. Palaeogeography, Palaeoclimatology, Palaeoecology, 291(3–4), 286–296. https://doi.org/10.1016/j.palaeo.2010.03.001
- Mateo, M. A., Romero, J., Pérez, M., Littler, M. M., & Littler, D. S. (1997). Dynamics of millenary organic deposits resulting from the growth of the Mediterranean seagrass *Posidonia oceanica. Estuarine, Coastal and Shelf Science*, 44(1), 103–110. https://doi.org/10.1006/ecss.1996.0116 Mazarrasa, I., Samper-Villarreal, J., Serrano, O., Lavery, P. S., Lovelock, C. E., Marbà, N., et al. (2018). Habitat characteristics provide insights of
- carbon storage in seagrass meadows. *Marine Pollution Bulletin*, 134(January), 106–117. https://doi.org/10.1016/j.marpolbul.2018.01.059 Meier, H. E. M., Broman, B., & Kjellström, E. (2004). Simulated sea level in past and future climates of the Baltic Sea. *Climate Research*, 27(1), 59–75. https://doi.org/10.3354/cr027059
- Moksnes, P., Eriander, L., Infantes, E., & Holmer, M. (2018). Local regime shifts prevent natural recovery and restoration of lost eelgrass beds along the Swedish west coast. *Estuaries and Coasts*, 41(6), 1712–1731. https://doi.org/10.1007/s12237-018-0382-y
- Moksnes, P., Gipperth, L., Eriander, L., Laas, K., Cole, S., & Infantes, E. (2016). Handbok för restaurering av ålgräs i Sverige: Vägledning. Havsoch vattenmyndigheten.
- Moksnes, P., Gullström, M., Tryman, K., & Baden, S. (2008). Trophic cascades in a temperate seagrass community. *Oikos*, 117(5), 763–777. https://doi.org/10.1111/j.2008.0030-1299.16521.x
- Moksnes, P., Röhr, E., Holmer, M., Eklöf, J., Eriander, L., Infantes, E., & Boström, C. (2021). Major impacts and societal costs of seagrass loss on sediment carbon and nitrogen stocks. *Ecosphere*, 12(7). https://doi.org/10.1002/ecs2.3658
- Molnár, M., Janovics, R., Major, I., Orsovszki, J., Gönczi, R., Veres, M., et al. (2013). Status report of the new AMS ¹⁴C sample preparation lab of the Hertelendi Laboratory of Environmental Studies (Debrecen, Hungary). *Radiocarbon*, 55(2), 665–676. https://doi.org/10.2458/azu_js_rc.55. 16394
- Moore, K. A., Shields, E. C., Parrish, D. B., & Orth, R. J. (2012). Eelgrass survival in two contrasting systems: Role of turbidity and summer water temperatures. *Marine Ecology Progress Series*, 448, 247–258. https://doi.org/10.3354/meps09578
- Mörner, N.-A. (1969). The Late Quaternary history of the Kattegatt Sea and the Swedish West Coast: Deglaciation, shorelevel displacement, chronology, isostasy and eustasy. Sveriges geologiska undersökning.
- Nejrup, L. B., & Pedersen, M. F. (2008). Effects of salinity and water temperature on the ecological performance of Zostera marina. Aquatic Botany, 88(3), 239–246. https://doi.org/10.1016/j.aquabot.2007.10.006
- Neukom, R., Steiger, N., Gómez-Navarro, J. J., Wang, J., & Werner, J. P. (2019). No evidence for globally coherent warm and cold periods over the preindustrial Common Era. *Nature*, 571(7766), 550–554. https://doi.org/10.1038/s41586-019-1401-2
- Ni, S., Quintana Krupinski, N. B., Chonewicz, J., Groeneveld, J., Knudsen, K. L., Seidenkrantz, M. S., & Filipsson, H. L. (2021). Seasonal climate variations in the Baltic Sea during the Last Interglacial based on foraminiferal geochemistry. *Quaternary Science Reviews*, 272, 107220. https:// doi.org/10.1016/j.quascirev.2021.107220
- Nielsen, S. L., Sand-Jensen, K., Borum, J., & Geertz-Hansen, O. (2002). Depth colonization of eelgrass (Zostera marina) and macroalgae as determined by water transparency in Danish coastal waters. Estuaries, 25(5), 1025–1032. https://doi.org/10.1007/BF02691349
- Olesen, B., & Sand-Jensen, K. (1994). Patch dynamics of eelgrass Zostera marina. Marine Ecology-Progress Series, 106, 147–156. https://doi. org/10.3354/meps106147
- Olsson, I. U. (1980). Content of ¹⁴C in marine mammals from northern Europe. *Radiocarbon*, 22(3), 662–675. https://doi.org/10.1017/s0033822200010031
- O'Neill, B. C., Tebaldi, C., Van Vuuren, D. P., Eyring, V., Friedlingstein, P., Hurtt, G., et al. (2016). The scenario model intercomparison project (ScenarioMIP) for CMIP6. *Geoscientific Model Development*, 9(9), 3461–3482. https://doi.org/10.5194/gmd-9-3461-2016
- Oreska, M. P. J., Wilkinson, G. M., McGlathery, K. J., Bost, M., & McKee, B. A. (2018). Non-seagrass carbon contributions to seagrass sediment blue carbon. *Limnology and Oceanography*, 63(S1), S3–S18. https://doi.org/10.1002/lno.10718
- Orth, R. J., Carruthers, T. I. M. J. B., Dennison, W. C., Carlos, M., Fourqurean, J. W., Jr, K. L. H., et al. (2006). A global crisis for seagrass ecosystems. *BioScience*, 56(12), 987–996. https://doi.org/10.1641/0006-3568(2006)56[987:agcfse]2.0.co;2
- Orth, R. J., Lefcheck, J. S., Mcglathery, K. S., Aoki, L., Luckenbach, M. W., Moore, K. A., et al. (2020). Restoration of seagrass habitat leads to rapid recovery of coastal ecosystem services (pp. 1–10).
- Orth, R. J., Marion, S. R., Moore, K. A., & Wilcox, D. J. (2010). Eelgrass (Zostera marina L.) in the Chesapeake Bay region of mid-Atlantic coast of the USA: Challenges in conservation and restoration. *Estuaries and Coasts*, 33(1), 139–150. https://doi.org/10.1007/s12237-009-9234-0
- Påsse, T., & Andersson, L. (2005). Shore-level displacement in fennoscandia calculated from empirical data. Gff, 127(4), 253–268. https://doi.org/ 10.1080/11035890501274253
- Påsse, T., & Daniels, J. (2015). Past shore-level and sea-level displacements. Sveriges geologiska undersökning.
- Petersen, C. G. J. (1893). Det videnskabelige udbytte af kanonbaaden "Hauchs" togter i de danske have indenfor Skagen, i aarene 1883-86 (chef: Premierlieutenant CF Drechsel) Atlas.
- Piñeiro-Juncal, N., Leiva-Dueñas, C., Serrano, O., Mateo, M. Á., & Martínez-Cortizas, A. (2020). Pedogenic processes in a Posidonia oceanica mat. Soil Systems, 4(18), 1–15. https://doi.org/10.3390/soilsystems4020018
- Pörtner, H.-O., Roberts, D. C., Adams, H., Adler, C., Aldunce, P., Ali, E., et al. (2022). Climate change 2022: Impacts, adaptation and vulnerability. IPCC Geneva.
- Poynter, J., & Eglinton, G. (1990). 14. Molecular composition of three sediments from hole 717c: The Bengal fan. Proceedings of the Ocean Drilling Program: Scientific results, 116, 155–161.
- Prentice, C., Hessing-Lewis, M., Sanders-Smith, R., & Salomon, A. K. (2019). Reduced water motion enhances organic carbon stocks in temperate eelgrass meadows. *Limnology and Oceanography*, 64(6), 1–16. https://doi.org/10.1002/lno.11191
- Prentice, C., Poppe, K. L., Lutz, M., Murray, E., Stephens, T. A., Spooner, A., et al. (2020). A synthesis of blue carbon stocks, sources and accumulation rates in eelgrass (*Zostera marina*) meadows in the Northeast Pacific. *Global Biogeochemical Cycles*, 34(2), e2019GB006345. https://doi.org/10.1029/2019gb006345
- Rabalais, N. N., Turner, R. E., Di, R. J., & Justic, D. (2009). Global change and eutrophication of coastal waters. ICES Journal of Marine Science, 66(7), 1528–1537. https://doi.org/10.1093/icesjms/fsp047
- Rasmusson, L. M., Buapet, P., George, R., Gullström, M., Gunnarsson, P. C. B., & Björk, M. (2020). Effects of temperature and hypoxia on respiration, photorespiration, and photosynthesis of seagrass leaves from contrasting temperature regimes. *ICES Journal of Marine Science*, 77(6), 2056–2065. https://doi.org/10.1093/icesjms/fsaa093
- Rasmusson, L. M., Gullström, M., Gunnarsson, P. C. B., George, R., & Björk, M. (2019). Estimation of a whole plant Q10 to assess seagrass productivity during temperature shifts. *Scientific Reports*, 9(1), 1–9. https://doi.org/10.1038/s41598-019-49184-z
- Rogala, J. T. (1997). Estimating fetch for navigation pools in the upper Mississippi river using a geographic information system.

- Rosenbauer, R. J., Grossman, E., & Takesue, R. (2006). A multi-index biomarker approach to understanding the paleo-occurrence of eelgrass (*Zostera marina*) in the nearshore of Puget Sound, Washington. In *Extended abstracts from the coastal habitats in puget sound (CHIPS) 2006 workshop* (p. 51).
- Rosenberg, R., Cato, I., Förlin, L., Grip, K., & Rodhe, J. (1996). Marine environment quality assessment of the Skagerrak-Kattegat. Journal of Sea Research, 35(1–3), 1–8. https://doi.org/10.1016/s1385-1101(96)90730-3
- Rosentau, A., Klemann, V., Bennike, O., Steffen, H., Wehr, J., Latinović, M., et al. (2021). A Holocene relative sea-level database for the Baltic Sea. *Quaternary Science Reviews*, 266, 107071. https://doi.org/10.1016/j.quascirev.2021.107071
- Salo, T., Pedersen, M. F., & Boström, C. (2014). Population specific salinity tolerance in eelgrass (Zostera marina). Journal of Experimental Marine Biology and Ecology, 461, 425–429. https://doi.org/10.1016/j.jembe.2014.09.010
- Samper-Villarreal, J., Lovelock, C. E., Saunders, M. I., Roelfsema, C., & Mumby, P. J. (2016). Organic carbon in seagrass sediments is influenced by seagrass canopy complexity, turbidity, wave height, and water depth. *Limnology and Oceanography*, 61(3), 938–952. https://doi.org/10. 1002/lno.10262
- Sanchez-Cabeza, J. A., Masqué, P., & Ani-Ragolta, I. (1998). ²¹⁰Pb and ²¹⁰Po analysis in sediments and soils by microwave acid digestion. Journal of Radioanalytical and Nuclear Chemistry, 227(1-2), 19-22. https://doi.org/10.1007/bf02386425
- Serrano, O., Davis, G., Lavery, P. S., Duarte, C. M., Martinez-Cortizas, A., Mateo, M. A., et al. (2016). Reconstruction of centennial-scale fluxes of chemical elements in the Australian coastal environment using seagrass archives. *Science of the Total Environment*, 541(September 2015), 883–894. https://doi.org/10.1016/j.scitotenv.2015.09.017
- Serrano, O., Lavery, P. S., Bongiovanni, J., & Duarte, C. M. (2020). Impact of seagrass establishment, industrialization and coastal infrastructure on seagrass biogeochemical sinks. *Marine Environmental Research*, 160(August 2019), 104990. https://doi.org/10.1016/j.marenvres.2020. 104990
- Serrano, O., Mateo, M. A., Dueñas-Bohórquez, A., Renom, P., López-Sáez, J. A., & Martínez Cortizas, A. (2011). The Posidonia oceanica marine sedimentary record: A Holocene archive of heavy metal pollution. Science of the Total Environment, 409(22), 4831–4840. https://doi.org/10. 1016/j.scitotenv.2011.08.001
- Short, F. T., Carruthers, T., Dennison, W., & Waycott, M. (2007). Global seagrass distribution and diversity: A bioregional model. Journal of Experimental Marine Biology and Ecology, 350(1-2), 3-20. https://doi.org/10.1016/j.jembe.2007.06.012
- Short, F. T., Davis, R. C., Kopp, B. S., Short, C. A., & Burdick, D. M. (2002). Site-selection model for optimal transplantation of eelgrass Zostera marina in the northeastern US. Marine Ecology Progress Series, 227, 253–267. https://doi.org/10.3354/meps227253
- Short, F. T., & Neckles, H. A. (1999). The effects of global climate change on seagrasses. Aquatic Botany, 63(3–4), 169–196. https://doi.org/10. 1016/S0304-3770(98)00117-X
 Siles, F. L. Uhla, M. F. Naddar, S. D. & Haward, M. F. (2000). Sources of organic matter in a specific matter in a specific matter in a specific matter in a specific matter.
- Sikes, E. L., Uhle, M. E., Nodder, S. D., & Howard, M. E. (2009). Sources of organic matter in a coastal marine environment: Evidence from nalkanes and their δ¹³C distributions in the Hauraki Gulf, New Zealand. *Marine Chemistry*, 113(3–4), 149–163. https://doi.org/10.1016/j. marchem.2008.12.003
- Skåne County Administrative Board. (2016). Ålgräs i Skåne 2016 -Fältinventering och satellitbildstolkning.
- Stevens, A. W., & Lacy, J. R. (2012). The influence of wave energy and sediment transport on seagrass distribution. *Estuaries and Coasts*, 35(1), 92–108. https://doi.org/10.1007/s12237-011-9435-1
- Stroeven, A. P., Hättestrand, C., Kleman, J., Heyman, J., Fabel, D., Fredin, O., et al. (2016). Deglaciation of fennoscandia. *Quaternary Science Reviews*, 147, 91–121. https://doi.org/10.1016/j.quascirev.2015.09.016
- Svensson, R. J., Asplund, M. E., Eklöf, A., & Lindegrath, M. (2021). Rapport av tilläggsuppdrag fälttest av videosläde för övervakning av ålgräs (Zostera marina) på västkusten.
- Törnqvist, T. E., & Hijma, M. P. (2012). Links between early Holocene ice-sheet decay, sea-level rise and abrupt climate change. Nature Geoscience, 5(9), 601–606. https://doi.org/10.1038/ngeo1536
- Unsworth, R. K. F., Collier, C. J., Waycott, M., Mckenzie, L. J., & Cullen-Unsworth, L. C. (2015). A framework for the resilience of seagrass ecosystems. *Marine Pollution Bulletin*, 100(1), 34–46. https://doi.org/10.1016/j.marpolbul.2015.08.016
- van Katwijk, M. M., Bos, A. R., De Jonge, V. N., Hanssen, L., Hermus, D. C. R., & De Jong, D. J. (2009). Guidelines for seagrass restoration: Importance of habitat selection and donor population, spreading of risks, and ecosystem engineering effects. *Marine Pollution Bulletin*, 58(2), 179–188. https://doi.org/10.1016/j.marpolbul.2008.09.028
- van Katwijk, M. M., Marieke, M., Thorhaug, A., Marbà, N., Orth, R. J., Duarte, C. M., et al. (2016). Global analysis of seagrass restoration: The importance of large-scale planting. *Journal of Applied Ecology*, 53(2), 567–578. https://doi.org/10.1111/1365-2664.12562
- Weltje, G. J., & Tjallingii, R. (2008). Calibration of XRF core scanners for quantitative geochemical logging of sediment cores: Theory and application. *Earth and Planetary Science Letters*, 274(3–4), 423–438. https://doi.org/10.1016/j.epsl.2008.07.054
- Wentworth, C. K. (1922). A scale of grade and class terms for clastic sediments. *The Journal of Geology*, 30(5), 377–392. https://doi.org/10.1086/622910
- Xu, S., Zhou, Y., Wang, P., Wang, F., Zhang, X., & Gu, R. (2016). Salinity and temperature significantly influence seed germination, seedling establishment, and seedling growth of eelgrass Zostera marina L. PeerJ, 2016(11), 1–21. https://doi.org/10.7717/peerj.2697
- Zhang, Y. H., Yu, B., Liu, Y. C., Ma, W., Li, W. T., & Zhang, P. D. (2022). The influence of decreased salinity levels on the survival, growth and physiology of eelgrass Zostera marina. Marine Environmental Research, 182(October), 105787. https://doi.org/10.1016/j.marenvres.2022. 105787
- Zhao, K., Wulder, M. A., Hu, T., Bright, R., Wu, Q., Qin, H., et al. (2019). Detecting change-point, trend, and seasonality in satellite time series data to track abrupt changes and nonlinear dynamics: A Bayesian ensemble algorithm. *Remote Sensing of Environment*, 232, 111181. https:// doi.org/10.1016/j.rse.2019.04.034